Author's response to reviews

Title: Soluble egg antigen of Schistosoma hematobium induces HCV replication in PBMC`s from patients with chronic HCV infection.

Authors:

Mostafa M El Awady (mkawady@yahoo.com)
Samar S Youssef (youssefsamar@hotmail.com)
Moataza H Omran (moataza13@yahoo.com)
Ashraf A Tabll (ashraftabll@yahoo.com)
Wael T El Garf (elgarf@yahoo.com)
Ahmed M Salem (ahmedsalem47@hotmail.com)

Version: 6 Date: 17 March 2006

Author's response to reviews:

a statement of The approval of the National Research center ethical medical committee to perform the research was included in the methods section